

G OPEN ACCESS

Citation: Argaw A, Toe LC, Hanley-Cook G, Dailey-Chwalibóg T, de Kok B, Ouédraogo L, et al. (2023) Effect of prenatal micronutrient-fortified balanced energy-protein supplementation on maternal and newborn body composition: A sub-study from the MISAME-III randomized controlled efficacy trial in rural Burkina Faso. PLoS Med 20(7): e1004242. https://doi.org/10.1371/journal.pmed.1004242

Received: February 9, 2023

Accepted: May 17, 2023

Published: July 24, 2023

Copyright: © 2023 Argaw et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The informed consent form does not allow sharing of personal data outside the research team. Request to access data need to be directed to the ethics committee of Ghent University Hospital through ethisch. <u>comite@uzgent.be</u>. Supporting study documents, including the study protocol and questionnaires, are publicly available on the study's website: https://misame3.ugent.be.

Funding: This work was supported by the Bill & Melinda Gates Foundation (Grant number: OPP1175213; https://www.gatesfoundation.org/)

RESEARCH ARTICLE

Effect of prenatal micronutrient-fortified balanced energy-protein supplementation on maternal and newborn body composition: A sub-study from the MISAME-III randomized controlled efficacy trial in rural Burkina Faso

Alemayehu Argaw¹, Laeticia Celine Toe^{1,2}, Giles Hanley-Cook¹, Trenton Dailey-Chwalibóg¹, Brenda de Kok¹, Lionel Ouédraogo³, Anderson Compaoré³, Moctar Ouédraogo³, Amadi Sawadogo³, Rasmané Ganaba³, Katrien Vanslambrouck¹, Patrick Kolsteren¹, Carl Lachat¹, Lieven Huybregts^{1,4}

1 Department of Food Technology, Safety and Health, Faculty of Bioscience Engineering, Ghent University, Ghent, Belgium, 2 Unité Nutrition et Maladies Métaboliques, Institut de Recherche en Sciences de la Santé (IRSS), Bobo-Dioulasso, Burkina Faso, 3 AFRICSanté, Bobo-Dioulasso, Burkina Faso, 4 Poverty, Health, and Nutrition Division, International Food Policy Research Institute (IFPRI), Washington, DC, United States of America

* Carl.Lachat@UGent.be

Abstract

Background

Micronutrient-fortified balanced energy-protein (BEP) supplements are promising interventions to prevent intrauterine growth retardation in low- and middle-income countries. On the other hand, one concern with blanket prenatal supplementation programs using energydense supplements is that they could lead to more maternal and/or infant overweight. However, evidence is lacking on the potential effect of BEP on maternal and offspring body composition. This study evaluates the effects of micronutrient-fortified BEP supplementation during pregnancy on body composition of mothers and their newborns in rural Burkina Faso.

Methods and findings

The MISAME-III study is an open label individually randomized controlled trial where pregnant women (n = 1,897) of gestational age <21 weeks received either a combination of micronutrient-fortified BEP and iron-folic acid (IFA) tablets (i.e., intervention) or IFA alone (i.e., control). The prenatal phase of the MISAME-III study was conducted between the first enrollment in October 2019 and the last delivery in August 2021. In a sub-study nested under the MISAME-III trial, we evaluated anthropometry and body composition in newborns who were born starting from 17 November 2020 (n: control = 368 and intervention = 352) and their mothers (n: control = 185 and intervention = 186). Primary study outcomes were newborn and maternal fat-free mass (FFMI) and fat-mass (FMI) indices. We used the deuterium dilution method to determine FFMI and FMI and %FFM and %FM of total body weight within 1 month postpartum. Our main analysis followed a modified intention-to-treat awarded to PK. The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: BEP, balanced energy-protein; BMI, body mass index; CI, confidence interval; CACE, complier average causal effect; D2O, deuterium oxide; EAR, estimated average requirement; FFMI, fat-free mass index; %FFM, fat-free mass as percentage of total body weight; FMI, fat-mass index; %FM, fat-mass as percentage of total body weight; FTIR, Fourier-transformed infrared; IFA, iron-folic acid; LBW, low birth weight; LNS, lipidbased nutrient supplement; MD, mean difference; MISAME, MIcronutriments pour la SAnté de la Mère et de l'Enfant; MMN, multimicronutrient tablet; MUAC, mid-upper arm circumference; RR, relative risk; SD, standard deviation; SGA, smallfor-gestational age, TBW, total body water. approach by analyzing all subjects with body composition data available. Univariable and multivariable linear regression models were fitted to compare the intervention and control arms, with adjusted models included baseline maternal age, height, arm fat index, hemoglobin concentration and primiparity, household size, wealth and food security indices, and newborn age (days). At study enrollment, the mean \pm SD maternal age was 24.8 \pm 6.13 years and body mass index (BMI) was 22.1 ± 3.02 kg/m² with 7.05% of the mothers were underweight and 11.5% were overweight. Prenatal micronutrient-fortified BEP supplementation resulted in a significantly higher FFMI in mothers (MD (mean difference): 0.45; 95% CI (confidence interval): 0.05, 0.84; P = 0.026) and newborns (MD: 0.28; 95% CI: 0.06, 0.50; P = 0.012), whereas no statistically significant effects were found on FMI. The effect of micronutrient-fortified BEP on maternal FFMI was greater among mothers from food secure households and among those with a better nutritional status (BMI >21.0 kg/m² or mid-upper arm circumference (MUAC) \geq 23 cm). Key limitations of the study are the relatively high degree of missing data (approximately 18%), the lack of baseline maternal body composition values, and the lack of follow-up body composition measurements to evaluate any longterm effects.

Conclusions

Micronutrient-fortified BEP supplementation during pregnancy can increase maternal and newborn FFMI, without significant effects on FMI.

Trial registration

ClinicalTrials.gov with identifier NCT03533712.

Author summary

Why was this study done?

- Prenatal multimicronutrient-fortified balanced energy-protein (BEP) supplementation is a promising nutritional intervention to alleviate the high burden of intrauterine growth retardation, including low birth weight and small-for-gestational age, in lowand middle-income countries.
- On the other hand, one concern with blanket prenatal supplementation programs using energy-dense supplements is that they could lead to more infant and/or maternal overweight, as observed with consumption of ultra-processed foods in high-income countries.
- However, only few studies evaluated the impact of prenatal BEP supplementation on the body composition of mothers and their newborns using a robust RCT design and a direct body composition assessment.

What did the researchers do and find?

- In a sub-study nested under the MISAME-III trial, an individually randomized controlled efficacy trial in rural Burkina Faso, we evaluated the effects of prenatal micronutrient-fortified BEP supplementation on body composition of mothers and their newborns.
- Our trial indicates that micronutrient-fortified BEP supplementation during pregnancy can improve maternal and newborn fat-free mass index (FFMI) without altering maternal fat-mass index (FMI).

What do these findings mean?

- Our findings that prenatal BEP improved FFMI without increases in FMI address the possible concern that prenatal BEP could lead to excess adiposity and associated long-term health risks.
- Future research is required to investigate the long-term effects of prenatal BEP on body composition of older infants and children.

Introduction

Low birth weight (LBW; i.e., born <2,500 g) is a marker of suboptimal fetal growth and development caused by preterm delivery (i.e., gestational duration <37 weeks) and/or fetal growth restriction, which results in a small-for-gestational age (SGA) newborn (i.e., birth weight below the tenth percentile of a reference group of newborns with the same gestational age). LBW is associated with a significantly higher neonatal mortality risk that varies between a 2-fold risk increment for SGA neonates, a 7-fold risk for preterm newborns, and up to a 15-fold risk increase when children are born both preterm and SGA [1]. Furthermore, LBW is associated with subsequent poor infant and child growth, morbidity, suboptimal early development, and adult chronic disease [2–4]. In 2015, 14.6% of newborns suffered from LBW as compared to 17.5% in 2000 with more than 90% of all LBW occurring in low- and middle-income countries [5]. Without additional efforts, it is unlikely that the World Health Assembly nutrition target to reduce LBW by 30% between 2012 and 2025 will be achieved.

Prenatal nutritional supplementation is one key strategy to prevent LBW. Systematic review and meta-analysis of relevant studies showed that prenatal multimicronutrient (MMN) fortified balanced energy-protein (BEP) supplementation reduces the risk of SGA (relative risk (RR): 0.79; 95% confidence interval (CI): 0.69, 0.90) [6], while prenatal food distribution was associated with an apparent reduction in LBW (RR: 0.92; 95% CI: 0.84, 1.0) [7]. One particular variety of MMN-fortified BEP supplements, known as lipid-based nutritional supplements (LNS), consists of promising food matrix to deliver the necessary macro- and micronutrients during gestation. LNS are typically in the form of peanut-based spreads that have low water activity, which prevents microbial proliferation, and are conveniently ready-to-consume. Two individually randomized controlled efficacy trials in Burkina Faso (i.e., MISAME-II and MIS-AME-III) assessed the effect of prenatal LNS on birth outcomes. Results of these trials suggest small to modest effects on birth length (MD (mean difference): 4.6 mm; 95% CI: 1.8, 7.3) in MISAME-II [8], and on birth weight (MD: 49.7 g; 95% CI: 10.8, 88.7) and birth length (MD: 2.0 mm; 95% CI: 0.1, 3.9) in MISAME-III [9]. In addition, the MISAME-III trial only found a marginal effect of BEP on gestational weight gain (0.28 kg; 95% CI: -0.05, 0.58 kg) [10].

One concern with blanket prenatal supplementation programs using energy-dense supplements, such as LNS, is that it could lead to more infant and/or maternal overweight, as observed with consumption of ultra-processed foods in high-income countries [11]. A clinical sub-study nested in the MISAME-II trial comparing prenatal LNS to MMN supplements found that LNS led to a higher cord blood leptin concentration [12]. Leptin in cord blood originates primarily from the fetal compartment and is a marker of neonatal fat mass [13,14]. The observed effect on cord leptin concentration was exacerbated in women with a higher body mass index (BMI) or mid-upper arm circumference (MUAC) at early gestation and was associated with a positive effect on birth weight. To further investigate this exploratory finding, we designed a sub-study nested in the main MISAME-III trial with as primary objectives (i) to assess if prenatal LNS altered neonatal body composition; and (ii) to examine if maternal nutritional status at early gestation modifies any effect of prenatal LNS on neonatal body composition. In addition, to address possible concerns related to prenatal BEP and maternal overweight or adiposity, we also assessed if prenatal LNS led to an effect on postpartum maternal body composition.

Methods

Ethics statement

The MISAME-III study protocol was approved by the ethics committee of Centre Muraz in Burkina Faso (N°2018–22/MS/SG/CM/CEI) and Ghent University Hospital in Belgium (B670201734334). Participant mothers provided their written consent for participation after an information session detailing the study, voluntary participation, and withdrawal. The trial is registered at ClinicalTrials.gov with registration number NCT03533712.

Study design and intervention

The MISAME-III study is an individually randomized 2×2 factorial trial evaluating the efficacy of a combination of micronutrient-fortified BEP and iron-folic acid (IFA) supplementation, as compared to IFA supplementation alone, during pregnancy and during lactation on various maternal, newborn and infant outcomes. In this sub-study nested in the main MISA-ME-III trial, we evaluate the effect of prenatal micronutrient-fortified BEP supplementation on maternal and newborn body composition.

The MISAME-III parent study has been described in detail previously [9,10,15]. In brief, the study was implemented in 6 rural health center catchment areas in the district of Houndé in the Hauts-Bassins region of Burkina Faso. The study area is characterized by a Sudano-Sahelian climate with a dry season running between September/October and April, and agricultural activities being the main livelihood of the community. The habitual diet during pregnancy is nondiverse [16], predominantly based on maize with a complement of leafy vegetables [17]. The mean daily energy intake from the base diet (i.e., excluding supplements) during pregnancy was estimated to be approximately 1,940 kcal in a subsample of MISA-ME-III participants at the end of the preharvest season, and dietary micronutrient intakes were inadequate to cover the estimated average requirements (EARs) [18]. During the previous MISAME-II trial, baseline prevalence of maternal underweight (BMI <18.5 kg/m²) during pregnancy was 12.9% [8].

Women were randomly assigned to receive either daily BEP and IFA supplementation (intervention) or IFA alone (control). The BEP supplement was a large-quantity LNS in the form of an energy-dense peanut paste fortified with multiple micronutrients [9]. A daily dose of the BEP (72 g) consisting of 36% lipids, 20% protein, and 32% carbohydrates provided an energy top-up of 393 kcal (S1 Table). The supplement covered at least the EARs of pregnant women for 11 micronutrients [19]. A daily dose of an IFA tablet (Sidhaant Life Sciences, Delhi, India) contained 65 mg iron [form: FeH₂O₅S] and 400 µg folic acid [form: $C_{19}H_{19}N_7O_6$]. Supplementation was provided under direct observation by locally trained village-based project workers conducting daily home visits of study participants.

Study participants

The prenatal phase of the main MISAME-III study was conducted between the first enrollment on 30 October 2019 and the last delivery on 7 August 2021. Pregnant women (n = 1,897) aged between 15 and 40 years with a gestational age less than 21 completed weeks were enrolled into the study. Women who reported an allergy to peanuts, who planned to leave or deliver outside of the study area, and who had multifetal pregnancy were excluded from the study. Data collection for the body composition sub-study was started on 17 November 2020, enrolling the 880 newborns delivered from that date onwards. From this sample, a random subsample of 390 mothers enrolled in this study. The random selection of mothers was generated by the Survey Solutions tablet-assisted personal interviewing software.

Stable isotope administration and saliva sample collection

The body composition of mothers and infants was assessed using the stable isotope dilution technique with deuterium oxide (D_2O) [20]. Mother-child pairs were invited to the health center during the third week postpartum. If the mother was absent at the scheduled visit, project village workers would conduct a home visit to invite the mother to attend 1 week later. In total, 3 saliva samples were collected from each subject using small cotton balls. First, a first baseline saliva sample was collected followed by an administration of the deuterium tracer. A pre-weighed constant dose of 1 g (dose volume of 0.9 mL) and 6 g (dose volume of 5.4 mL) of D₂O (99.8%, Sigma-Aldrich, St. Louis, Canada) was administered to infants and mothers, respectively. The doses of D₂O were weighed in a laboratory in Bobo-Dioulasso, to the nearest 0.0001 g, using a precision balance (model SAB 224i, Adams Equipment, Felde, Germany). Then, a second saliva sample was collected 2 h and 30 min after administering the tracer for children and 3 h for mothers. Lastly, a third saliva sample was collected 30 min after the second sample. Each neonatal and maternal saliva sample was approximately 1.5 to 2 mL in volume. For newborns, the cotton balls were wrapped around a straw and saliva was collected by swabbing the inside of the neonate's cheeks. Mothers were asked to chew on the cotton balls to moisten them. Cotton balls were expressed in 20 mL disposable syringes and the saliva was collected in a 2 mL disposable cryovial. The cryovials were labeled and stored in a zip lock bag in a portable cooler (2 to 6°C). After completion of the sample collection, samples were transported to the project office in Houndé and stored in a dedicated freezer at -18° C. Samples were transported in cool cryoboxes from the project office to the laboratory in Bobo-Dioulasso once every week where they were stored at -20° C until the time of analysis.

In the event that both the mother and her newborn were selected (n = 390) for the substudy, the D₂O dose for the mother was administered after the collection of the last neonatal saliva sample to avoid any transfer of D₂O tracer from the mother to the newborn through breastmilk. Breastmilk was expressed using a medical-grade breast pump (Medela, Mississauga, Canada) into sterile baby bottles, labeled and stored in a cool box. Mothers were allowed to bottle-feed their child during the 2.5 h equilibration phase (between the second and first saliva sample). The net weight of the milk consumed was recorded using scales with a precision of 1 g (model KD7000, <u>www.myweigh.com</u>). The field experiment and procedures were pilot tested in 12 mother-infant dyads.

The D₂O concentrations in saliva samples were measured by Fourier-transformed infrared (FTIR) spectrophotometry using an Agilent 4500 Series device (Agilent, California, United States of America). All samples were measured in duplicate. The mean of the 4 measurements of the post-administration saliva samples was used for analysis. A calibration curve was compiled prior to every measurement session to convert the signal into a D₂O concentration. The isotopic enrichment of saliva was calculated by subtracting the quantity of D₂O measured in the baseline sample from that measured in the post-dose samples. The D₂O dilution volume was calculated by dividing the administered dose of D₂O by the isotopic enrichment. Total body water (TBW) was calculated by multiplying the D₂O dilution volume by 1.044 to account for 4.4% proton loss. For newborns who were breastfed between sample collections, we subtracted the weighed quantity of consumed breastmilk multiplied by 87%, which is the estimated water content of breastmilk [21], from the TBW. Fat-free mass (FFM) in kg was calculated from TBW using age- and sex-specific hydration factors for newborns [22] and mothers [23]. Fat mass (FM) was calculated by subtracting the FFM from total body weight.

Study outcomes and measurement

The primary outcomes of interest of the study are fat-mass index (FMI) and fat-free mass index (FFMI) (i.e., mass divided by squared height) in mothers and newborns. Secondary study outcomes are %FFM and %FM of total body weight in mothers and newborns; maternal BMI; and newborn length-for-age, weight-for-age, and weight-for-length z-scores. A sample size of 440 newborns per study arm (880 in total) was required to detect an effect size (Cohen's d) of 0.2 on FFMI and 0.4 for an interaction effect between supplementation and maternal low versus high BMI (<21 versus $\geq 21 \text{ kg/m}^2$) or MUAC (<23 versus $\geq 23 \text{ cm}$) with a type I error of 5%, a statistical power of 80%, and assuming 10% sample loss. Furthermore, a sample size of 195 women per study arm (390 in total) was required to detect an effect size (Cohen's d) of 0.3 on FFMI with a type I error of 5%, a statistical power of 5%, a statistical power of 80%, and assuming 10% sample loss.

Body composition assessment of mothers and newborns was conducted within 3 weeks postpartum (mean ± SD days after delivery: 18.2 ± 3.96). Newborn weight was measured to the nearest 10 g using a Seca 384 scale, and length was measured to the nearest 1 mm using a Seca 416 length board. Maternal weight was measured to the nearest 100 g using a Seca 876 scale and height was measured to the nearest 1 mm using a ShorrBoard. All measurements were taken in duplicate and a third measurement was taken when there was a large discrepancy between the duplicate measurements. We used the average of the 2 closest anthropometric measurements for analysis. Anthropometric indices of length-for-age, weight-for-age, and weight-for-length *z*scores were calculated based on the World Health Organization 2006 Child Growth Standards [24]. Household food security status was evaluated using Household Food Insecurity Access Scale [25], and the UNICEF/WHO water and sanitation tool was employed to assess access to improved drinking water and improved sanitation facilities [26]. Household wealth status was assessed by a wealth index score generated using the principal components analysis based on availability of household assets, facilities, and housing conditions.

Statistical analysis

Data analyses were conducted using Stata 17.1 (Statacorp, Texas, USA) and a two-sided statistical significance was considered at alpha <0.05. Descriptive statistics are presented as means \pm SD for the continuous variables and as frequencies (%) for the categorical variables. Study outcomes were checked for normality of distribution using visual inspection of histograms and Q-Q plots of the outcome values and their residual terms.

Unadjusted and adjusted group differences in body composition indices were estimated by fitting linear regression models. All models included health center and randomization block as covariates to account for the study design [15]. Adjusted regression models additionally included prespecified known prognostic factors of the outcomes and factors that are associated with body composition indices and that differ between study arms by >2.5%. These included maternal age, height, arm fat index, hemoglobin concentration and primiparity at study enrollment, household size, wealth index and food security index, and the number of postpartum days before the body composition measurement. The main analysis followed a modified intention-to-treat approach by analyzing all participants having body composition measurements. Outcome values that were found to be extreme (i.e., standardized values >abs[3 z-scores]) were inspected individually and excluded from data analysis if deemed implausible. We also conducted a per-protocol analysis by comparing only participants with a BEP adherence of at least 75% with the control subjects. Following a suggestion of one reviewer of the manuscript, we deviated from our statistical analysis plan replacing the naïve per-protocol analysis by a more robust complier average causal effect (CACE) analysis. The CACE approach has advantage over the per-protocol analysis in maintaining the balance between treatment groups obtained through the randomization process by comparing compliers in the BEP group versus "inferred compliers" (would be compliers) in the control group [27]. For this purpose, using the gsem command in Stata [28], we applied a latent class regression modeling approach simultaneously fitting models for compliance and for the intervention effect. Compliance status was predicted by covariates such as household food insecurity, access to improved water and sanitation, number of under-five children, and number of job activities by the women and the household head, while the effects of BEP on the study outcomes were estimated by adjusting for the aforementioned covariates.

Besides evaluating any effect modification of BEP by maternal BMI and MUAC at study enrollment, we explored the presence of effect modification by other covariates. For this purpose, we tested an interaction effect of the intervention and each factor on FFMI and FMI. Effect modification by a subgroup factor was considered in the presence of a statistically significant interaction (P < 0.10). We also applied the approach by Katz and colleagues [29] to evaluate whether the treatment effect of prenatal BEP on FFMI and FMI was comparable between mothers and newborns across the maternal BMI distribution. In this approach, differences in maternal and newborn FFMI and FMI between intervention and control groups are estimated as nonlinear smooth functions of the percentiles of maternal BMI distribution at baseline. Lastly, we fit linear regression models to evaluate the relationships between maternal and newborn FFMI, and FMI and FMI and FMI and birth anthropometry.

Results

From the 880 newborns considered for the body composition sub-study, we analyzed data from 720 (81.8%) newborns (**Fig 1**). One hundred and sixty newborns were not included in the analysis due to a lack of sample collection or lab analyses not being conducted (n = 148) or implausible values in anthropometric or body composition outcomes (n = 12). We analyzed data from 371 (95.1%) of the 390 mothers selected for the current sub-study, whereas the remaining 19 mothers were not included in our analysis due to a lack of sample collection or lab analyses not being conducted (n = 17) or implausible anthropometric or body composition values (n = 2). Table 1 presents the characteristics of participants in the newborn and maternal





https://doi.org/10.1371/journal.pmed.1004242.g001

body composition sub-studies. Study groups were comparable with regard to baseline household and maternal characteristics in both the newborn and maternal body composition study samples. Participant characteristics in the newborn and maternal body composition sub-studies were also similar to those from the main MISAME-III study (S2 Table). Mean \pm SD maternal age in the newborn subsample was 24.8 \pm 6.13 years with 41.7% of participants having had 3 or more pregnancies and 58.6% were from food-insecure households. At enrollment, during the first and early second trimester of pregnancy, mean \pm SD maternal BMI, MUAC, and arm fat index were 22.1 \pm 3.02 kg/m², 26.2 \pm 2.64 cm, and 25.5 \pm 7.26, respectively, and 7.05% of the mothers were underweight (BMI <18.5 kg/m²) and 11.5% were overweight (BMI \geq 25.0 kg/m²). We observed a high adherence of mothers to both the BEP (84.2%) and IFA (89.3%) supplements, with compliance to IFA supplement being comparable between control (88.4%) and intervention (90.3%) groups.

Prenatal micronutrient-fortified BEP supplementation resulted in a significantly higher FFMI among mothers (adjusted MD: 0.45; 95% CI: 0.05, 0.84; P = 0.026) and newborns (adjusted MD: 0.28; 95% CI: 0.06, 0.50; P = 0.012) (Table 2). However, we did not find statistically significant group differences in other maternal and newborn body composition variables including FMI, %FFM, and %FM, as well as in anthropometric indices such as maternal BMI

Characteristics	Infa	nt sample	Maternal sample		
	Control (<i>n</i> = 440)	Intervention (<i>n</i> = 440)	Control $(n = 195)$ Intervention $(n = 195)$		
Health center catchment area					
Boni	110 (25.0)	102 (23.2)	45 (23.1)	49 (25.1)	
Dohoun	38 (8.60)	44 (10.0)	21 (10.8)	25 (12.8)	
Dougoumato II	88 (20.0)	82 (18.6)	34 (17.4)	32 (16.4)	
Karaba	42 (9.50)	47 (10.7)	21 (10.8)	16 (8.20)	
Kari	84 (19.1)	93 (21.1)	35 (17.9)	34 (17.4)	
Koumbia	78 (17.7)	72 (16.4)	39 (20.0)	39 (20.0)	
Household level					
Wealth index, 0 to 10 points	4.53 ± 1.72	4.74 ± 1.76	4.51 ± 1.69	4.69 ± 1.74	
Household food insecurity ²	257 (58.4)	260 (59.1)	119 (61.0)	110 (56.4)	
Improved primary water source ³	268 (60.9)	273 (62.0)	106 (54.4)	122 (62.6)	
Improved sanitation facility ⁴	260 (59.1)	241 (54.8)	109 (55.9)	112 (57.4)	
Household size	6.44 ± 4.52	6.47 ± 4.62	6.49 ± 4.59	6.63 ± 4.82	
Polygamous households	129 (29.3)	129 (29.3)	62 (31.8)	55 (28.2)	
Maternal					
Age, years	25.1 ± 6.11	24.9 ± 6.11	25.1 ± 5.99	24.7 ± 5.97	
Ethnic group					
Bwaba	261 (59.3)	254 (57.7)	115 (59.0)	104 (53.3)	
Mossi	148 (33.6)	149 (33.9)	74 (37.9)	72 (36.9)	
Others	31 (7.05)	37 (8.41)	6 (3.08)	19 (9.74)	
Religion					
Muslim	178 (40.5)	185 (42.0)	82 (42.1)	92 (47.2)	
Animist	105 (23.9)	99 (22.5)	44 (22.6)	36 (18.5)	
Protestant	88 (20.0)	86 (19.5)	36 (18.5)	37 (19.0)	
Catholic	56 (12.7)	54 (12.3)	30 (15.4)	26 (13.3)	
No religion, no animist	13 (3.00)	14 (3.20)	3 (1.50)	4 (2.10)	
Primary education and above	189 (45.0)	181 (41.1)	90 (46.2)	89 (45.6)	
Gestational age, weeks	11.0 ± 3.87	10.9 ± 3.86	10.9 ± 3.86	10.7 ± 3.84	
Trimester of gestation					
First	295 (67.0)	299 (68.0)	136 (69.7)	132 (67.7)	
Second	145 (33.0)	141 (32.0)	59 (30.3)	63 (32.3)	
Parity					
0	90 (20.5)	102 (23.2)	37 (19.0)	45 (23.1)	
1-2	166 (37.7)	143 (32.5)	77 (39.5)	62 (31.8)	
3 or more	184 (41.8)	195 (44.3)	81 (41.5)	88 (45.1)	
Weight, kg	58.8 ± 9.09	58.6 ± 8.80	58.3 ± 7.39	58.9 ± 9.47	
Height, cm	163 ± 5.92	163 ± 5.95	163 ± 5.96	163 ± 5.80	
BMI, kg/m ²	22.2 ± 3.00	22.1 ± 2.95	21.9 ± 2.49	22.2 ± 3.21	
MUAC, mm	263 ± 26.9	262 ± 25.4	261 ± 22.3	262 ± 25.9	
Subscapular skinfold thickness, mm	12.1 ± 5.83	12.0 ± 5.70	11.7 ± 4.53	12.0 ± 5.63	
Fripital skinfold thickness, mm	11.7 ± 4.73	11.9 ± 4.74	11.6 ± 4.27	12.0 ± 5.10	
Arm muscle area, cm ^{2,5}	40.8 ± 6.75	40.6 ± 6.58	40.2 ± 5.76	40.4 ± 5.75	
Arm fat area, cm ^{2,5}	14.6 ± 7.21	14.8 ± 6.92	14.1 ± 5.51	14.7 ± 7.17	
Arm fat index ⁵	25.3 ± 7.22	25.9 ± 7.36	25.4 ± 6.82	25.8 ± 7.51	
Newborn					
Sex, female	195 (47.1)	207 (51.0)	91 (47.2)	93 (48.2)	

Table 1. Participant characteristics between study arms in the newborn and maternal body composition study¹.

(Continued)

Table 1. (Continued)

Characteristics	Infa	nt sample	Maternal sample			
	Control (<i>n</i> = 440)	Intervention $(n = 440)$	Control (<i>n</i> = 195)	Intervention $(n = 195)$		
Age, days	18.1 ± 3.69	18.0 ± 4.14	18.0 ± 3.83	17.9 ± 3.80		

¹Values are frequencies (percentages) or means ± SDs.

²Assessed using FANTA/USAID's Household Food Insecurity Access Scale.

³Protected well, borehole, pipe, or bottled water were considered improved water sources.

⁴Flush toilet connected to local sewage or septic tank, or pit latrine with slab and/or ventilation were considered improved sanitation facilities.

⁵Arm muscle area = (mid-upper arm circumference* π)²/(4* π); arm fat area = ((mid-upper arm circumference)²/(4* π))—arm muscle area; arm fat index = arm fat area/ ((mid-upper arm circumference)²/(4* π)).

BMI, body mass index; MUAC, mid-upper arm circumference; SD, standard deviation.

https://doi.org/10.1371/journal.pmed.1004242.t001

and newborn length-for-age, weight-for-age, and weight-for-length *z*-scores. In CACE analysis including only participants with a BEP adherence of at least 75%, we found that prenatal BEP additionally resulted in significantly higher %FFM in mothers (adjusted MD: 1.44; 95% CI: 0.15, 2.73; P = 0.029) and newborns (adjusted MD: 2.31; 95% CI: 0.72, 3.89; P = 0.004), and lower %FM in mothers (adjusted MD: -1.44; 95% CI: -2.73, -0.15; P = 0.029) and newborns (adjusted MD: -1.44; 95% CI: -2.73, -0.15; P = 0.029) and newborns (adjusted MD: -1.44; 95% CI: -2.73, -0.15; P = 0.029) and newborns (adjusted MD: -1.44; 95% CI: -2.73, -0.15; P = 0.029) and newborns (adjusted MD: -1.44; 95% CI: -2.73, -0.15; P = 0.029) and newborns (adjusted MD: -1.44; 95% CI: -2.73, -0.15; P = 0.029) and newborns (adjusted MD: -1.44; 95% CI: -2.73, -0.15; P = 0.029) and newborns (adjusted MD: -2.31; 95% CI: -3.89, -0.72; P = 0.004) (S3 Table).

Characteristics	Control	Intervention	Unadjusted difference (95% CI)	Р	Adjusted difference (95% CI)	Р
Maternal (n)	185	186				
FFMI (kg/m ²)	17.5 ± 1.78	17.9 ± 2.10	0.50 (0.09, 0.90)	0.015	0.45 (0.05, 0.84)	0.026
FMI (kg/m2)	5.14 ± 2.23	4.99 ± 2.18	-0.15 (-0.59, 0.29)	0.496	-0.22 (-0.58, 0.14)	0.237
%FFM	77.7 ± 7.87	78.6 ± 7.41	0.93 (-0.57, 2.43)	0.223	1.08 (-0.22, 2.39)	0.103
%FM	22.3 ± 7.87	21.4 ± 7.41	-0.93 (-2.43, 0.57)	0.223	-1.08 (-2.39, 0.22)	0.103
BMI (kg/m ²)	22.6 ± 2.29	22.7 ± 2.42	0.11 (-0.38, 0.61)	0.653	0.09 (-0.32, 0.51)	0.657
Newborn (n)	368	352				
FFMI (kg/m ²)	12.6 ± 1.53	12.9 ± 1.57	0.29 (0.07, 0.51)	0.011	0.28 (0.06, 0.50)	0.012
FMI (kg/m ²)	0.95 ± 1.33	0.86 ± 1.30	-0.15 (-0.33, 0.04)	0.122	-0.13 (-0.32, 0.05)	0.146
%FFM	93.2 ± 9.41	93.9 ± 9.37	1.05 (-0.27, 2.37)	0.119	0.97 (-0.33, 2.28)	0.143
%FM	6.78 ± 9.41	6.11 ± 9.37	-1.05 (-2.37, 0.27)	0.119	-0.97 (-2.28, 0.33)	0.143
Length-for-age z-score	-0.70 ± 1.11	-0.62 ± 0.99	0.08 (-0.07, 0.23)	0.287	0.07 (-0.07, 0.21)	0.327
Weight-for-age z-score	-0.56 ± 0.95	-0.44 ± 0.90	0.10 (-0.03, 0.24)	0.121	0.11 (-0.02, 0.23)	0.094
Weight-for-length z-score	-0.24 ± 1.10	-0.12 ± 1.03	0.12 (-0.03, 0.27)	0.105	0.13 (-0.02, 0.27)	0.080

 1 Values are means ± SDs. Unadjusted and adjusted group differences were estimated by fitting linear regression models. All models contained the health center and randomization block as covariates to take into account clustering of our data by the study design. Adjusted models contained additional covariates such as maternal age, height, arm fat index, hemoglobin concentration and parity at study enrollment, household size, wealth index and food insecurity status, and number of postpartum days before the body composition measurement.

BEP, balanced energy-protein

BMI, body mass index

CI, confidence interval

FFMI, fat-free mass index

%FFM, fat-free mass as percentage of total body weight

FMI, fat-mass index

%FM, fat-mass as percentage of total body weight

SD, standard deviation.

https://doi.org/10.1371/journal.pmed.1004242.t002

Subgroup factor		Control		Intervention	Unadjusted difference (95% CI)	Р	Adjusted difference (95% CI)	Р
	n	Mean ± SD	n	Mean ± SD				
Maternal								
Household food security						0.108		0.038
Food secure	76	17.5 ± 1.65	81	18.3 ± 2.20	0.87 (0.13, 1.62)	0.022	1.07 (0.29, 1.86)	0.008
Food insecure	109	17.5 ± 1.88	105	17.6 ± 1.97	0.40 (-0.16, 0.96)	0.157	0.23 (-0.34, 0.81)	0.427
Maternal BMI						0.014		0.010
\geq 21.0 kg/m ²	110	17.9 ± 1.83	116	18.7 ± 2.11	0.93 (0.36, 1.50)	0.002	0.88 (0.29, 1.47)	0.004
<21.0 kg/m ²	75	16.9 ± 1.53	70	16.6 ± 1.27	-0.29 (-0.82, 0.23)	0.271	-0.47 (-1.03, 0.09)	0.099
Maternal MUAC						0.006		0.004
≥23 cm	82	17.8 ± 1.98	76	18.9 ± 2.25	1.52 (0.77, 2.27)	< 0.001	1.54 (0.74, 2.34)	< 0.001
<23 cm	103	17.2 ± 1.56	110	17.2 ± 1.67	0.02 (-0.44, 0.49)	0.924	-0.15 (-0.61, 0.31)	0.530
Newborn								
Household food security						0.498		0.496
Food secure	150	12.6 ± 1.55	147	12.9 ± 1.68	0.38 (-0.01, 0.78)	0.059	0.31 (-0.10, 0.71)	0.135
Food insecure	221	12.7 ± 1.60	206	12.9 ± 1.64	0.27 (-0.05, 0.59)	0.103	0.27 (-0.06, 0.59)	0.108
Maternal BMI						0.452		0.285
\geq 21.0 kg/m ²	230	12.8 ± 1.53	222	13.0 ± 1.64	0.18 (-0.12, 0.48)	0.243	0.14 (-0.17, 0.44)	0.371
<21.0 kg/m ²	141	12.4 ± 1.62	132	12.7 ± 1.66	0.40 (-0.01, 0.81)	0.057	0.48 (0.07, 0.90)	0.023
Maternal MUAC						0.736		0.892
≥23 cm	170	12.7 ± 1.59	145	13.0 ± 1.75	0.30 (-0.08, 0.68)	0.117	0.24 (-0.14, 0.62)	0.218
<23 cm	201	12.6 ± 1.57	209	12.8 ± 1.58	0.24 (-0.08, 0.55)	0.137	0.25 (-0.07, 0.57)	0.122

Table 3. Subgroup analyses of the efficacy of	prenatal micronutrient-fortified BEP supplementation on maternal and newborn FFMI ¹ .
	F

¹Linear regression models were fitted to test the interaction between study arm and a subgroup factor at P < 0.10. All models contained the health center and randomization block to account for clustering by the study design. Adjusted models contained additional covariates such as maternal age, height, arm fat index, hemoglobin concentration and parity at study enrollment, household size, wealth index and food security status, and number of postpartum days before the body composition measurement.

BEP, balanced energy-protein

BMI, body mass index

CI, confidence interval

FFMI, fat-free mass index

MUAC, mid-upper arm circumference

SD, standard deviation.

https://doi.org/10.1371/journal.pmed.1004242.t003

In a subgroup analysis, the positive effect of prenatal BEP on maternal FFMI was found among mothers who belonged to food-secure households (adjusted MD: 1.07; 95% CI: 0.29, 1.86; P = 0.008; $P_{\text{interaction}} = 0.038$) and who had higher BMI (BMI $\geq 21.0 \text{ kg/m}^2$) (adjusted MD: 0.88; 95% CI: 0.29, 1.47; P = 0.004; $P_{\text{interaction}} = 0.010$) and MUAC (MUAC $\geq 23 \text{ cm}$) (adjusted MD: 1.54; 95% CI: 0.74, 2.34; P < 0.001; $P_{\text{interaction}} = 0.004$) at study inclusion (**Table 3**). We did not observe effect modification of BEP supplementation by any of the above variables on maternal FMI or newborn FFMI and FMI (**Table 4**).

When exploring the treatment effects on maternal FFMI by maternal BMI percentile at study inclusion, we observed a larger BEP treatment effect at higher percentiles of the maternal BMI distribution (S1 Fig). A more complex pattern emerged for newborn FFMI with stronger BEP effects at low and high BMI percentiles. The treatment effects of BEP on maternal and newborn FMI across maternal BMI percentiles showed a less consistent pattern (S2 Fig). The relationship between maternal and newborn FFMI also showed no evidence of a tradeoff between newborn and mother of the BEP treatment effect on FFMI (S3 Fig). Lastly, we found

Subgroup factor		Control		Intervention	Unadjusted difference (95% CI)	P	Adjusted difference (95% CI)	P
	n	Mean ± SD	n	Mean ± SD				
Maternal								
Household food security						0.720		0.892
Food secure	76	5.39 ± 2.45	81	5.13 ± 2.37	-0.29 (-1.21, 0.63)	0.536	-0.12 (-0.87, 0.64)	0.758
Food insecure	109	4.96 ± 2.07	105	4.88 ± 2.03	-0.27 (-0.86, 0.31)	0.354	-0.30 (-0.83, 0.23)	0.272
Maternal BMI						0.277		0.112
\geq 21.0 kg/m ²	110	5.91 ± 2.40	116	5.53 ± 2.44	-0.58 (-1.28, 0.13)	0.107	-0.59 (-1.21, 0.03)	0.063
<21.0 kg/m ²	75	4.00 ± 1.31	70	4.08 ± 1.20	0.14 (-0.34, 0.62)	0.570	0.11 (-0.38, 0.61)	0.652
Maternal MUAC						0.981		0.726
≥23 cm	82	6.11 ± 2.69	76	6.05 ± 2.54	-0.46 (-1.48, 0.56)	0.371	-0.61 (-1.54, 0.32)	0.198
<23 cm	103	4.36 ± 1.39	110	4.25 ± 1.51	-0.10 (-0.51, 0.32)	0.649	-0.11 (-0.46, 0.25)	0.552
Newborn								
Household food security						0.643		0.783
Food secure	149	0.95 ± 1.31	146	0.77 ± 1.28	-0.19 (-0.51, 0.13)	0.249	-0.12 (-0.44, 0.20)	0.457
Food insecure	221	0.90 ± 1.44	205	0.90 ± 1.38	-0.11 (-0.38, 0.17)	0.438	-0.12 (-0.39, 0.16)	0.406
Maternal BMI						0.581		0.906
\geq 21.0 kg/m ²	229	0.95 ± 1.42	221	0.87 ± 1.31	-0.03 (-0.29, 0.22)	0.805	-0.04 (-0.30, 0.21)	0.742
<21.0 kg/m ²	141	0.88 ± 1.34	131	0.81 ± 1.38	-0.11 (-0.46, 0.22)	0.509	-0.05 (-0.40, 0.29)	0.760
Maternal MUAC						0.951		0.753
≥23 cm	169	1.06 ± 1.40	143	0.99 ± 1.39	-0.09 (-0.42, 0.23)	0.568	-0.09 (-0.42, 0.24)	0.590
<23 cm	201	0.81 ± 1.37	209	0.75 ± 1.29	-0.08 (-0.34, 0.19)	0.572	-0.05 (-0.31, 0.22)	0.730

Table 4. Subgroup analyses of the efficacy of prenatal micronutrient-fortified BEP supplementation on maternal and newborn FMI¹.

¹Linear regression models were fitted to test interaction between study arm and a subgroup factor at P < 0.10. All models contained the health center and randomization block to account for clustering by the study design. Adjusted models contained additional covariates such as maternal age, height, arm fat index, hemoglobin concentration and parity at study enrollment, household size, wealth index and food security status, and number of postpartum days before the body composition measurement.

BEP, balanced energy-protein

BMI, body mass index

CI, confidence interval

FMI, fat-mass index

MUAC, mid-upper arm circumference

SD, standard deviation.

https://doi.org/10.1371/journal.pmed.1004242.t004

a positive relationship between maternal and newborn FFMI and FMI and birth anthropometry, such as birth weight, MUAC, and ponderal index (<u>S4 Table</u>).

Discussion

Our study found that prenatal BEP led to a higher maternal and newborn FFMI without significant effects on FMI. The positive effect of prenatal BEP on maternal FFMI was found to be larger among mothers with a better nutritional status at baseline (i.e., BMI \geq 21.0 kg/m² or MUAC \geq 23 cm).

Our results are in contrast to those from the MISAME-II trial conducted in the same health district between 2006 and 2008. MISAME-II found elevated cord blood leptin concentration, indicating larger newborn FM accretion [13,14], in the group who received prenatal MMN-fortified LNS supplementation as compared to MMN alone [12]. Conversely, our findings suggest that instead of pathways favoring FM accretion, the small positive effects of BEP observed on newborn size at birth [9] are driven by other physiologic and metabolic mechanisms. These

may include increased nutrient availability to the placenta [30,31], improvements in maternal hematologic profile [32], favorable maternal immune responses and related lower burden of infectious diseases, asymptomatic inflammation and oxidative stress [31,33,34], or other epigenetic gene regulation pathways [31,35,36]. A study in a cohort of Kenyan women and their newborns also found positive associations between newborn birth weight and maternal TBW and FFM, but not with FM, assessed using isotope dilution technique [37]. On the other hand, a cohort study in pregnant women from an urban slum in India reported that maternal FFM in the third trimester was negatively associated with LBW, while FM was negatively associated with the occurrence of SGA [38].

A number of trials have evaluated the effects of prenatal LNS on maternal and newborn outcomes [6,39–42], but data on newborn body composition are lacking. A prenatal LNS supplementation made from peanut butter (400 kcal/day) provided to a small sample of primigravid women in Pakistan did not affect postpartum maternal body composition, using bioelectrical impedance scale, as compared to the control supplement made from wheat flakes and skimmed milk (138 kcal/day) [43]. Another trial evaluated the effects of an energy-protein food supplement provided starting either from early (approximately 9 weeks of gestation) or later pregnancy (approximately 20 weeks of gestation), in combination with IFA or MMN supplements, in a large-sample of women in Bangladesh. The study found that offspring body composition using a bioelectrical impedance scale at the age 54 months was not affected by the prenatal supplementation [44]. Similarly, LNS supplementation provided to mothers during pregnancy and lactation and continued in their children aged 6 to 18 months did not result in any long-term effect on child FFM and FM at the age of 4 to 6 years as compared to control supplements containing IFA and calcium or MMN [45].

Various possible explanations can be forwarded for our finding that prenatal BEP in mothers and their newborns led to FFM accretion rather than FM accretion. First, the composition of the BEP supplement used could possibly influence the balance between FFM and FM tissue accretions. Diets and supplements containing animal-source proteins, especially those from dairy products, support lean mass accretion better than diets or supplements containing only plant-based protein sources [46]. Unlike the LNS supplement used in MISAME-II, which contained only plant-based protein sources (soy and peanut protein) with a protein digestibility corrected amino acid score of 0.9, the BEP in the current study contained both plant- and animal-source protein (i.e., soy: 61%, peanut: 15%, and milk: 25%) with an amino acid score of 1.1. Furthermore, the BEP provided a range of micronutrients that are essential for the synthesis of maternal and fetal tissues [47], which can be reflected in improved FFM accretion. Contrary to the current study where control mothers were given IFA, MISAME-II provided MMN to the control group which may have supported FFM accretion and thereby reduced any relative impact of the LNS on FFM. Second, the effects of prenatal BEP on body composition could depend on maternal baseline nutritional status and energy balance during pregnancy. Extra energy intake is required during the second and third trimester of gestation to support adequate gestational weight gain and the increase in basal metabolic rate [48]. In the absence of an adequate energy balance, FFM preservation in essential maternal and fetal tissue compartments could be prioritized over FM preservation. Given the lack of impact of BEP on maternal fat mass in mothers with higher BMI at baseline, we hypothesize that the additional energy provided by the BEP may have been insufficient to fill the additional energy requirement due to pregnancy to result in higher postpartum fat mass. A 24 h recall survey in a subsample of the study participants found that the median energy intake, including BEP, was 2,329 kcal/day during the lean season [18], which is lower than our estimated energy requirement of mothers. The mean daily energy requirement based on maternal weight and a physical activity level of 1.79 (estimated in MISAME-II mothers during the lean season [49]) was

estimated at 2,430 kcal without including the additional daily requirements of 290 kcal and 465 kcal during the second and third trimester of pregnancy [50]. A previous dietary recall survey in the same health district also revealed that pregnant women did not have a higher energy intake than non-pregnant women, which suggests that the additional energy requirement due to pregnancy was not accounted for by their dietary intake [17].

Subgroup analyses showed that prenatal BEP was more efficacious in increasing maternal FFM index in the women from food secure households or with a more adequate energy balance (i.e., baseline BMI \geq 21.0 kg/m² or MUAC \geq 23 cm). These findings are in line with our previous finding that BEP was more efficacious in reducing SGA prevalence among mothers with a more adequate baseline nutritional status (e.g., nonanemic, higher MUAC) [9]. One possible explanation for the lack of significant effect on FFM among women with low BMI could be due to a limited amino acids supply from endogenous breakdown to meet the demands of pregnancy. A study in a small sample of low BMI Indian women showed that BEP supplementation using 300 kcal/d and 15 g protein/d starting from gestational age of 12 weeks did not result in improved whole-body protein synthesis, improved fluxes in the methyl group precursors serine and glycine, or better pregnancy outcomes including higher gestational weight gain and newborn weight [51]. These findings might suggest extending BEP supplementation to the pre-conception period to leverage the efficacy of prenatal BEP on optimal maternal and newborn outcomes.

The observed effects of prenatal BEP on newborn FFMI can be seen in relation to the subsequent modest effect of prenatal BEP on linear growth achieved at 6 months of age (0.11 z-score; 95% CI: 0.01, 0.21; p = 0.032) that we reported previously [52]. In Ethiopia, Admassu and colleagues [53] found that newborn FFM, but not FM, was positively associated with height-for-age z-score at 2 years of age. Furthermore, FFM in the first month of life was associated with linear growth at the age of 1 year [54]. A Danish cohort also found that newborns with more FFM were on average both taller and heavier at 3 years of age [55].

A major strength of our study was the direct measurement of body composition using the deuterium dilution method, which enables robust assessment that can be implemented at scale in a rural field setting. We were also able to achieve a high adherence to both the BEP and IFA supplements through daily home visits by community-based study workers who observed supplement intake. A dietary intake survey conducted in a subsample of the study participants also ruled out any substitution effect of the BEP on usual dietary patterns [18]. The CACE analysis evaluating the intervention's effects among subjects with better compliance also confirmed our findings in the main analysis.

The study has several limitations that need to be addressed. About 18% of the newborn outcome data was missing mainly due to saliva samples not being collected or unreadable identification numbers on cryotubes, whereas we anticipated a 10% loss when designing the study. However, the occurrence of missing data was not specific to any of the study group and maternal and household characteristics were comparable between study groups. Therefore, we argue that the missing data are unlikely to have biased group comparisons. Maternal body composition using the deuterium dilution method was not measured at baseline. As such, we were only able to consider baseline adjustment in our models for proxy variables of body composition such as BMI and arm fat index. Lastly, having follow-up body composition measurements would have revealed whether observed effects are sustained in the long term.

In conclusion, our study demonstrated that prenatal BEP supplementation has positive effects on newborn and maternal FFM accretion without significant alteration of FMI. The absence of any effects of BEP on fat mass accretion in both newborn and mother addresses the possible concern that prenatal BEP would lead to excess adiposity and associated long-term health risks.

Supporting information

S1 Table. Nutritional values of the ready-to-use supplementary food for pregnant women. (DOCX)

S2 Table. Comparison of participants between the MISAME-III study and the body composition sub-studies.

(DOCX)

S3 Table. Effect of prenatal micronutrient-fortified BEP supplementation on maternal and newborn body composition using CACE estimates. (DOCX)

S4 Table. Relationship between newborn and maternal FFMI and FMI and birth anthropometry.

(DOCX)

S1 Fig. Treatment effect on maternal and newborn FFMI across the distribution of maternal BMI at baseline. The estimated difference in FFMI between the intervention and control groups is plotted as a function of the percentiles of maternal BMI. The zero line indicates no efficacy of BEP. The positive y values indicate a higher FFMI in the intervention group, and the negative y values indicate a lower FFMI, with upper and lower 95% confidence bands. BMI, body mass index; FFMI, fat-free mass index. (TIF)

S2 Fig. Treatment effect on maternal and newborn FMI across the distribution of maternal BMI at baseline. The estimated difference in FMI between the intervention and control groups is plotted as a function of the percentiles of maternal BMI. The zero line indicates no efficacy of BEP. The positive y values indicate a higher FMI in the intervention group, and the negative y values indicate a lower FMI, with upper and lower 95% confidence bands. BMI, body mass index; FFI, fat-mass index. (TIF)

S3 Fig. Locally weighted scatterplot smoothing plotting the relationship between newborn and maternal FFMI. Linear regression model was fitted to estimate the beta and 95% CIs for the relationship between newborn and maternal FFMI. (TIF)

S1 CONSORT Checklist. CONSORT checklist of the manuscript. (PDF)

S1 Statistical Analysis Plan. Impact of a prenatal and postnatal balanced energy protein supplement on birth size and postnatal child growth in Burkina Faso. (PDF)

Acknowledgments

The authors thank all the women from Boni, Dohoun, Karaba, Dougoumato II, Koumbia, and Kari who participated in the study, the data collection team and Henri Somé from AFRIC-Santé. We thank Nutriset (France) for donating the BEP supplements.

Author Contributions

Conceptualization: Laeticia Celine Toe, Patrick Kolsteren, Carl Lachat, Lieven Huybregts.

Data curation: Alemayehu Argaw, Lionel Ouédraogo, Lieven Huybregts.

Formal analysis: Alemayehu Argaw, Lieven Huybregts.

Funding acquisition: Patrick Kolsteren, Carl Lachat.

- Investigation: Alemayehu Argaw, Laeticia Celine Toe, Giles Hanley-Cook, Trenton Dailey-Chwalibóg, Brenda de Kok, Anderson Compaoré, Moctar Ouédraogo, Amadi Sawadogo, Rasmané Ganaba, Katrien Vanslambrouck.
- Methodology: Laeticia Celine Toe, Brenda de Kok, Moctar Ouédraogo, Katrien Vanslambrouck, Patrick Kolsteren, Lieven Huybregts.
- **Project administration:** Laeticia Celine Toe, Brenda de Kok, Moctar Ouédraogo, Rasmané Ganaba, Katrien Vanslambrouck, Patrick Kolsteren, Carl Lachat.

Resources: Laeticia Celine Toe, Brenda de Kok, Moctar Ouédraogo, Katrien Vanslambrouck.

Software: Alemayehu Argaw, Giles Hanley-Cook, Brenda de Kok, Katrien Vanslambrouck, Lieven Huybregts.

Supervision: Patrick Kolsteren, Carl Lachat, Lieven Huybregts.

Validation: Laeticia Celine Toe, Patrick Kolsteren, Carl Lachat, Lieven Huybregts.

Visualization: Alemayehu Argaw, Trenton Dailey-Chwalibóg.

Writing - original draft: Alemayehu Argaw, Lieven Huybregts.

Writing – review & editing: Laeticia Celine Toe, Giles Hanley-Cook, Trenton Dailey-Chwalibóg, Brenda de Kok, Patrick Kolsteren, Carl Lachat.

References

- Katz J, Lee ACC, Kozuki N, Lawn JE, Cousens S, Blencowe H, et al. Mortality risk in preterm and smallfor-gestational-age infants in low-income and middle-income countries: A pooled country analysis. Lancet. 2013; 382(9890):417–425. https://doi.org/10.1016/S0140-6736(13)60993-9 PMID: 23746775
- Gluckman PD, Hanson MA, Beedle AS. Early life events and their consequences for later disease: a life history and evolutionary perspective. Am J Hum Biol. 2007 Jan; 19(1):1–19. <u>https://doi.org/10.1002/ ajhb.20590 PMID: 17160980</u>
- Christian P, Lee SE, Angel MD, Adair LS, Arifeen SE, Ashorn P, et al. Risk of childhood undernutrition related to small-for-gestational age and preterm birth in low- and middle-income countries. Int J Epidemiol. 2013; 42(5):1340–1355. https://doi.org/10.1093/ije/dyt109 PMID: 23920141
- Sacchi C, Marino C, Nosarti C, Vieno A, Visentin S, Simonelli A. Association of intrauterine growth restriction and small for gestational age status with childhood cognitive outcomes: A systematic review and meta-analysis. JAMA Pediatr. 2020; 174(8):772–781. <u>https://doi.org/10.1001/jamapediatrics.2020</u>. 1097 PMID: 32453414
- Blencowe H, Krasevec J, de Onis M, Black RE, An X, Stevens GA, et al. National, regional, and worldwide estimates of low birthweight in 2015, with trends from 2000: a systematic analysis. Lancet Glob Health. 2019 Jul; 7(7):e849–60. https://doi.org/10.1016/S2214-109X(18)30565-5 PMID: 31103470
- Ota E, Hori H, Mori R, Tobe-Gai R, Farrar D. Antenatal dietary education and supplementation to increase energy and protein intake. Cochrane Database Syst Rev. 2015; 2015(6). <u>https://doi.org/10. 1002/14651858.CD000032.pub3 PMID: 26031211</u>
- Lassi ZS, Padhani ZA, Rabbani A, Rind F, Salam RA, Das JK, et al. Impact of dietary interventions during pregnancy on maternal, neonatal, and child outcomes in low-and middle-income countries. Nutrients. 2020; 12(2). https://doi.org/10.3390/nu12020531 PMID: 32092933
- Huybregts L, Roberfroid D, Lanou H, Menten J, Meda N, Van Camp J, et al. Prenatal food supplementation fortified with multiple micronutrients increases birth length: A randomized controlled trial in rural Burkina Faso. Am J Clin Nutr. 2009; 90(6):1593–1600. https://doi.org/10.3945/ajcn.2009.28253 PMID: 19812173

- de Kok B, Toe LC, Hanley-Cook G, Argaw A, Ouédraogo M, Compaoré A, et al. Prenatal fortified balanced energy-protein supplementation and birth outcomes in rural Burkina Faso: A randomized controlled efficacy trial. PLoS Med. 2022; 19(5):1–20.
- Hanley-Cook G, Toe LC, Tesfamariam K, de Kok B, Argaw A, Compaoré A, et al. Fortified Balanced Energy-Protein Supplementation, Maternal Anemia, and Gestational Weight Gain: A Randomized Controlled Efficacy Trial among Pregnant Women in Rural Burkina Faso. J Nutr. 2022; 152:2277–2286. https://doi.org/10.1093/jn/nxac171 PMID: 35906874
- Wang Y, Wang K, Du M, Khandpur N, Rossato SL, Lo CH, et al. Maternal consumption of ultra-processed foods and subsequent risk of offspring overweight or obesity: results from three prospective cohort studies. BMJ. 2022 Oct; 379:e071767. https://doi.org/10.1136/bmj-2022-071767 PMID: 36198411
- Huybregts L, Roberfroid D, Lanou H, Meda N, Taes Y, Valea I, et al. Prenatal lipid-based nutrient supplements increase cord leptin concentration in pregnant women from rural burkina faso. J Nutr. 2013; 143(5):576–583. https://doi.org/10.3945/jn.112.171181 PMID: 23535609
- 13. Clapp J, Kiess W. Cord blood leptin reflects fetal fat mass. J Soc Gynecol Investig. 1998; 5:300–303. PMID: 9824809
- Lepercq J, Challier JC, Guerre-Millo M, Cauzac M, Vidal H, Hauguel-De MS. Prenatal leptin production: Evidence that fetal adipose tissue produces leptin. J Clin Endocrinol Metab. 2001; 86(6):2409–2413. https://doi.org/10.1210/jcem.86.6.7529 PMID: 11397832
- Vanslambrouck K, De Kok B, Toe LC, De Cock N, Ouedraogo M, Dailey-Chwalibóg T, et al. Effect of balanced energy-protein supplementation during pregnancy and lactation on birth outcomes and infant growth in rural Burkina Faso: Study protocol for a randomised controlled trial. BMJ Open. 2021; 11 (3):1–10. https://doi.org/10.1136/bmjopen-2020-038393 PMID: 33762226
- Hanley-Cook GT, Argaw A, de Kok B, Toe LC, Dailey-Chwalibóg T, Ouédraogo M, et al. Seasonality and Day-to-Day Variability of Dietary Diversity: Longitudinal Study of Pregnant Women Enrolled in a Randomized Controlled Efficacy Trial in Rural Burkina Faso. J Nutr. 2022; 152(9):2145–2154. <u>https:// doi.org/10.1093/jn/nxac104 PMID: 35524695</u>
- Huybregts LF, Roberfroid DA, Kolsteren PW, Van Camp JH. Dietary behaviour, food and nutrient intake of pregnant women in a rural community in Burkina Faso. Matern Child Nutr. 2009; 5(3):211–222. https://doi.org/10.1111/j.1740-8709.2008.00180.x PMID: 20572925
- de Kok B, Argaw A, Hanley-Cook G, Toe LC, Ouédraogo M, Dailey-Chwalibóg T, et al. Fortified Balanced Energy-Protein Supplements Increase Nutrient Adequacy without Displacing Food Intake in Pregnant Women in Rural Burkina Faso. J Nutr. 2021; 151(12):3831–3840. <u>https://doi.org/10.1093/jn/ nxab289 PMID: 34494113</u>
- Food and Nutrition Board, Institute of Medicine, National Academies. Dietary Reference Intakes (DRIs): Estimated Average Requirements. 2011. Available from: https://www.ncbi.nlm.nih.gov/books/ NBK109829/. Accessed on 2022 Oct 10.
- Shepherd JA, Heymsfield SB, Norris SA, Redman LM, Ward LC, Slater C. Measuring Body Composition in Low-Resource Settings across the Life Course. Obesity. 2016; 24(5):985–988. <u>https://doi.org/10.1002/oby.21491</u> PMID: 27060932
- Holland B, Welch AA, Unwin ID, Buss DH, Paul AA, Southgate DAT. McCance and Widdowson's The Composition of Foods. 5th ed. Royal Society of Chemistry/Ministry of Agriculture, Fisheries and Food. 1991.
- Butte NF, Hopkinson JM, Wong WW, Smith EO, Ellis KJ. Body Composition during the First 2 Years of Life: An Updated Reference. Pediatr Res. 2000; 47(5):578–85. https://doi.org/10.1203/00006450-200005000-00004 PMID: 10813580
- 23. International Atomic Energy Agency. Introduction to body composition assessment using deuterium dilution technique with analysis of urine samples by isotope ratio mass spectrometry. Vienna; 2010. Available from: https://www.iaea.org/publications/8370/introduction-to-body-composition-assessment-using-the-deuterium-dilution-technique-with-analysis-of-urine-samples-by-isotope-ratio-mass-spectrometry. Accessed 2021 June 21.
- 24. de Onis M, Gaza C, Victoria C, Onyango A, Fragillo E, Martines J. The WHO Multicenter Growth Reference Study: planning, study design, and methodology. Food Nutr Bull. 2004; 25:S15–S26.
- Coates J, Swindale A, Bilinsky P. Household Food Insecurity Access Scale (HFIAS) for Measurement of Food Access: Indicator Guide. 2007. Available from: https://www.fantaproject.org/sites/default/files/ resources/HFIAS_ENG_v3_Aug07.pdf. Accessed 2020 Sep 11.
- World Health Organization & United Nations Children's Fund (UNICEF). Core questions on drinking water and sanitation for household surveys. World Health Organization. 2006. Available from: https://apps.who.int/iris/handle/10665/43489. Accessed 2021 Oct 2.

- Peugh JL, Strotman D, Mcgrady M, Rausch J, Kashikar-zuck S. Beyond intent to treat (ITT): A complier average causal effect (CACE) estimation primer. J Sch Psychol. 2017; 60:7–24. https://doi.org/10. 1016/j.jsp.2015.12.006 PMID: 28164801
- 28. Troncoso P, Morales-Gómez A. Estimating the complier average causal effect via a latent class approach using gsem. Stata J. 2022; 22(2):404–415.
- Katz J, Christian P, Dominici F, Zeger SL. Treatment effects of maternal micronutrient supplementation vary by percentiles of the birth weight distribution in rural Nepal. J Nutr. 2006; 136(5):1389–1394. https://doi.org/10.1093/jn/136.5.1389 PMID: 16614435
- Wu G, Bazer FW, Cudd TA, Meininger CJ, Spencer TE. Maternal Nutrition and Fetal Development. J Nutr. 2004; 134:2169–2172. https://doi.org/10.1093/jn/134.9.2169 PMID: 15333699
- Jansson T, Powell TL. Role of the placenta in fetal programming: Underlying mechanisms and potential interventional approaches. Clin Sci. 2007; 113(1–2):1–13. https://doi.org/10.1042/CS20060339 PMID: 17536998
- Fawzi WW, Msamanga GI, Kupka R, Spiegelman D, Villamor E, Mugusi F, et al. Multivitamin supplementation improves hematologic status in HIV-infected women and their children in Tanzania. Am J Clin Nutr. 2007; 85(5):1335–1343. https://doi.org/10.1093/ajcn/85.5.1335 PMID: 17490971
- Steketee RW. Pregnancy, nutrition and parasitic diseases. J Nutr. 2003; 133:1661S–1667S. <u>https://doi.org/10.1093/jn/133.5.1661S PMID: 12730482</u>
- Stein P, Scholl TO, Schluter MD, Leskiw MJ, Chen X, Spur BW, et al. Oxidative stress early in pregnancy and pregnancy outcome. Free Radic Res. 2008; 42:841–848. https://doi.org/10.1080/ 10715760802510069 PMID: 18985484
- Christian P, Stewart CP. Maternal micronutrient deficiency, fetal development, and the risk of chronic disease. J Nutr. 2010; 140(3):437–445. https://doi.org/10.3945/in.109.116327 PMID: 20071652
- Langley-Evans SC. Fetal programming of cardiovascular function through exposure to maternal undernutrition. Proc Nutr Soc. 2001; 60(4):505–513. https://doi.org/10.1079/pns2001111 PMID: 12069404
- Wanjohi MN, Ogada I, Wekesah FM, Khayeka-Wandabwa C, Kimani-Murage EW. Relationship between maternal body composition during pregnancy and infant's birth weight in Nairobi informal settlements. Kenya BMJ Nutr Prev Heal. 2020; 3(2):151–161.
- Mamidi RS, Banjara SK, Manchala S, Babu CK, Geddam JJB, Boiroju NK, et al. Maternal Nutrition, Body Composition and Gestational Weight Gain on Low Birth Weight and Small for Gestational Age- A Cohort Study in an Indian Urban Slum. Children. 2022; 9(10).
- Imdad A, Bhutta ZA. Maternal nutrition and birth outcomes: Effect of balanced protein-energy supplementation. Paediatr Perinat Epidemiol. 2012; 26(SUPPL. 1):178–190. <u>https://doi.org/10.1111/j.1365-3016.2012.01308.x PMID: 22742610</u>
- 40. Hambidge KM, Westcott JE, Garcés A, Figueroa L, Goudar SS, Dhaded SM, et al. A multicountry randomized controlled trial of comprehensive maternal nutrition supplementation initiated before conception: The Women First trial. Am J Clin Nutr. 2019; 109(2):457–469. https://doi.org/10.1093/ajcn/nqy228 PMID: 30721941
- 41. Bliznashka L, Sudfeld CR, Garba S, Guindo O, Soumana I, Adehossi I, et al. Prenatal supplementation with multiple micronutrient supplements or medium-quantity lipid-based nutrient supplements has limited effects on child growth up to 24 months in rural Niger: a secondary analysis of a cluster randomized trial. Am J Clin Nutr. 2021;1–11.
- Callaghan-Gillespie M, Schaffner AA, Garcia P, Fry J, Eckert R, Malek S, et al. Trial of ready-to-use supplemental food and corn-soy blend in pregnant Malawian women with moderate malnutrition: a randomized controlled clinical trial. Am J Clin Nutr. 2017 Oct; 106(4):1062–9. https://doi.org/10.3945/ajcn. 117.157198 PMID: 28793991
- **43.** Tariq K, Fatima S, Nazli R, Habib S. Impact of Lipid-Based Nutritional Supplementation in Primi-Gravidas and Its Effect on Maternal, Birth and Infant Outcomes: a randomized controlled trial. Khyber Med Univ J. 2021; 13(4):187–192.
- 44. Khan AI, Kabir I, Hawkesworth S, Ekström EC, Arifeen S, Frongillo EA, et al. Early invitation to food and/or multiple micronutrient supplementation in pregnancy does not affect body composition in offspring at 54 months: Follow-up of the MINIMat randomised trial. Bangladesh Matern Child Nutr. 2015; 11(3):385–397.
- 45. Kumordzie SM, Adu-Afarwuah S, Arimond M, Young RR, Adom T, Boatin R, et al. Maternal and infant lipid-based nutritional supplementation increases height of Ghanaian children at 4–6 years only if the mother was not overweight before conception. J Nutr. 2019; 149(5):847–855. https://doi.org/10.1093/jn/nxz005 PMID: 31034033

- 46. Lim MT, Pan BJ, Toh DWK, Sutanto CN, Kim JE. Animal protein versus plant protein in supporting lean mass and muscle strength: A systematic review and meta-analysis of randomized controlled trials. Nutrients. 2021; 13(2):1–18. https://doi.org/10.3390/nu13020661 PMID: 33670701
- Mousa A, Naqash A, Lim S. Macronutrient and micronutrient intake during pregnancy: An overview of recent evidence. Nutrients. 2019; 11(2):1–20. https://doi.org/10.3390/nu11020443 PMID: 30791647
- Butte NF, Wong WW, Treuth MS, Ellis KJ, Smith EOB. Energy requirements during pregnancy based on total energy expenditure and energy deposition. Am J Clin Nutr. 2003; 79:1078–1087.
- 49. Vandevijvere S, Kolsteren P, Van Camp J. Evaluatie van de voedselconsumptie en de fysieke activiteit van zwangere vrouwen in Houndé, Burkina Faso. 2007. Available from: <u>https://lib.ugent.be/catalog/ rug01:001230433</u>. Accessed on 2022 Sep 10.
- FAO, WHO, UNU. Human energy requirements: report of a Joint FAO/WHO/UNU Expert Consultation. Rome; 2001. Available from: http://www.fao.org/3/y5686e/y5686e0 8.htm. Accessed on 2022 Sep 5.
- Dwarkanath P, Hsu JW, Tang GJ, Anand P, Thomas T, Thomas A, et al. Energy and protein supplementation does not affect protein and amino acid kinetics or pregnancy outcomes in underweight Indian women. J Nutr. 2016; 146(2):218–226. https://doi.org/10.3945/jn.115.218776 PMID: 26764317
- 52. Argaw A, de Kok B, Toe LC, Hanley-Cook G, Dailey-Chwalibóg T, Ouédraogo M, et al. Fortified balanced energy-protein supplementation during pregnancy and lactation and infant growth in rural Burkina Faso: A 2 × 2 factorial individually randomized controlled trial. PLoS Med. 2023; 20(2):e1004186.
- Admassu B, Wells JCK, Girma T, Andersen GS, Owino V, Belachew T, et al. Body composition at birth and height at 2 years: A prospective cohort study among children in Jimma. Ethiopia Pediatr Res. 2017; 82(2):209–214.
- Admassu B, Ritz C, Wells JCK, Girma T, Andersen GS, Belachew T, et al. Accretion of fat-free mass rather than fat mass in infancy is positively associated with linear growth in childhood. J Nutr. 2018; 148 (4):607–615. https://doi.org/10.1093/jn/nxy003 PMID: 29659955
- 55. Berglund NR, Lewis JI, Michaelsen KF, Mølgaard C, Renault KM, Carlsen EM. Birthweight z-score and fat-free mass at birth predict body composition at 3 years in Danish children born from obese mothers. Acta Paediatr Int J Paediatr. 2022; 111(7):1427–1434.